


























Site 39 Remedial Action 
Status Update 

10/08/2008 
 

Documents 
 
Document Schedule Comments 

Habitat Restoration Plan Issue Final by 10/31 Received additional comments from 
DTSC/DF&G 

ROD Amendment Issued Draft Final on 9/17 Agency comments were due 10/6 
RD/RAWP Issue Draft by 10/17  
   
 
Ongoing Engineering Activities 
 
Site 39 

• Data management and reporting of HMP plant species survey (transects) for remediation 
areas. 

 
OU2 Landfills 

• Evaluate E/F Hill as borrow source for vegetative cover. Grading plan sent to Army on 
9/24. 

 
Future Field Activities 

• Wetland monitoring - Ponds 8 and 30 still need survey data if possible before range 
remediation. No ponding from 2006 through 2008. 

• Site Preparation scheduled to start January 2009. 
• Excavation scheduled to start March 2009. 

 
 



Thermal Treatment Unit
Operation Summary

2007/2008

4/4/2006
10/8/2008 16:30

6528
2891.50
44.3%
8760

4035.4
48.7%
6760.5
2189.4
32.4%
41.3%

Pounds of Methane Removed (2007) 372759
Pounds of Methane Removed (2008) 166694

Cumulative % Operation (since 4/4/2006):

% Operation (2007):
Total Hours (2008):
Total Hours Operated (2008):
% Operation (2008):

Start Date/Time:
Last Reading Date/Time:

Total Hours (2007):
Total Hours Operated (2007):

Total Hours (2006):
Total Hours Operated (2006):
% Operation (2006):

Methane Concentration vs. Time
(after 02-13-08)

Interior Extraction Wells
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note: EW-30 and EW-31 are not currently being pumped



OPERABLE UNIT CARBON TETRACHLORIDE PLUME 
A-AQUIFER REMEDIAL ACTION 

 
STATUS – October 14, 2008 

 
FIELD WORK 
 

• EISB pilot study system construction complete – October 25 
• EISB pilot study substrate injection initiated - January 29 
• 7000 gallons substrate injected – March 27 
• EISB pilot study system shut off – April 11 
• EISB pilot study monitoring completed – July 24 
• Equipment demobed from site – September 26 

 
SCHEDULE 

 
• Subsequent quarterly monitoring for EISB pilot study conducted under Groundwater 

Monitoring Program. 
• Draft RA Work Plan/RD (Attachment 1 – A-Aquifer) [Agency Comments Due] – 

November 17 
• Preston Park Sports Complex shut down November 17 through February 1 
 

DATA (Preliminary) 
 
• None 

 
PROBLEMS/CHANGES 
 

• None  



HGL AGENDA & NOTES 
 

Fort Ord HTW BCT Meeting 
1:00 PM, 14 October 2008 

Santa Rosa, California 
 
1. Groundwater Remediation System Update 
 
Northwest Treatment System (NWTS) has operated continuously since last update on 18 September 
2008.  Total volume pumped through 06 October 2008 is 68,804,400 gallons.  The average treatment 
rate over the last four weeks has been 90.9 gallons per minute. 

 
The last bi-monthly samples from the treatment system and extraction wells were collected on 29 
September and analytical results are not yet back from the laboratory. The validated data from the 
July 2008 sampling was reported previously and is summarized below:  
 

• System influent TCE concentration decreased slightly to 3.6 µg/L (from 3.9 µg/L in May). 
• System influent cis-1,2-DCE concentration increased slightly to 0.41µg/L (from 0.36 µg/L 

in May). 
• System effluent concentrations were below detection limit for all ten compounds 

monitored. 
• TCE at individual extraction wells (except MW-85) and system influent slightly lower 

than the previous sample in May; MW-85 increased from 2.5 to 4.4 ppb. 
• All extraction wells showed TCE concentrations less than 5 µg/L except the two wells 

located in the central portion of the plume: MW-87 (9.1 µg/L) and EW-71 (14 µg/L). 
 
 
2. Long Term Monitoring Update 
 
Third quarter (includes annual frequency wells) samples were collected during the period from 29 
September through 02 October.  The bi-monthly system performance samples were collected on 29 
September.  
 
 
3. NWTS Operation Relative to Carbon Change-out 
 
HGL submitted a letter to the US Army that summarized the agreement reached at the September 
BCT meeting concerning the criteria for carbon change-out based on Cis-1,2-DCE concentrations in 
the performance sample routinely collected at the mid-point of the GAC process (i.e., as influent to 
the lag vessels).  As agreed, the carbon will be changed if the TCE concentration in the mid-point 
sample exceeds 0.5 µg/L or if the cis-1,2-DCE concentration exceeds 3.0 µg/L. 
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